Author's response to reviews

Title: Effect of Sunlight Exposure on Cognitive Function Among Depressed and Non-Depressed Participants: a REGARDS Cross-Sectional Study

Authors:

Shia T Kent (shia@uab.edu)
Leslie A McClure (lmclure@uab.edu)
William L Crosson (bill.crosson@nasa.gov)
Donna K Arnett (Arnett@ms.soph.uab.edu)
Virginia G Wadley (vwadley@uab.edu)
Nalini Sathiakumar (NSathiakumar@ms.soph.uab.edu)

Version: 6 Date: 13 May 2009

Author's response to reviews:

Dear EH Editorial Team,

The following has been completed as requested:

- On the title page, format the email addresses in paragraph format with the email addresses separated with semi-colons.
- Remove the page numbers.
- Include a List of Abbreviations before the Competing Interests section.
- Remove the issue numbers in the References section.
- The tables should all be in closed cell format so that the data and headings are clearly separated by single line borders.

Please contact with any additional clarifications/requests.

Thanks,
Shia Kent